ECONSTOR Make Your Publications Visible.

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Halla, Martin; Wagner, Alexander F.; Zweimüller, Josef

Working Paper Does immigration into their neighborhoods incline voters toward the extreme right? The case of the freedom Party of Austria

Working Paper, No. 83

Provided in Cooperation with:

Department of Economics, University of Zurich

Suggested Citation: Halla, Martin; Wagner, Alexander F.; Zweimüller, Josef (2012) : Does immigration into their neighborhoods incline voters toward the extreme right? The case of the freedom Party of Austria, Working Paper, No. 83, University of Zurich, Department of Economics, Zurich, https://doi.org/10.5167/uzh-63504

This Version is available at: https://hdl.handle.net/10419/77527

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU

Does Immigration Into Their Neighborhoods Incline Voters Toward the Extreme Right? The Case of the Freedom Party of Austria

Martin Halla^{*} Alexander F. Wagner[†] Josef Zweimüller[‡]

July 1, 2012

Abstract

This paper explores one potentially important channel through which immigration may drive support for extreme-right-wing parties: the presence of immigrants in the voters' neighborhoods. We study the case of the Freedom Party of Austria (FPÖ). Under the leadership of Jörg Haider, this party increased its share of votes from less than 5 percent in the early 1980s to 27 percent by the year 1999. Using past regional settlement patterns as a source of exogenous variation, we find a significantly positive effect on FPÖ votes of the residential proximity of immigrants and citizens, explaining roughly a quarter of the cross-community variance in those votes. It is the proximity of low- and medium-skilled immigrants that drives this result; high-skilled immigrants have no (or even a negative) effect on FPÖ votes.

JEL Classification: P16, J61. Keywords: Immigration, political economy, voting.

^{*}University of Linz and IZA; email: martin.halla@jku.at

[†]University of Zurich, CEPR, and Harvard Kennedy School; email: alexander.wagner@bf.uzh.ch.

[‡]Corresponding Author, University of Zurich, CEPR, CESifo, and IZA. Address: Department of Economics, University of Zurich, Mühlebachstrasse 86, CH-8008 Zurich, Switzerland; email: josef.zweimueller@econ.uzh.ch

1. Introduction

Voters in many European countries — including Austria, Belgium, Denmark, France, Netherlands, Norway, and Switzerland — have recently expressed strong support for extreme-rightwing (ERW) parties. From the 1970s until the mid-1980s, hardly any ERW party had gained more than five percent in a general election. Fifteen years later, some ERW parties in the above-mentioned countries received between ten and twenty-five percent of the votes. History reminds us that the rise of extreme parties within a democratic environment can put democracy itself at risk (Almond and Verba, 1965; Dahl, 1989). Although few political movements today are direct analogues of the *National Socialist German Workers' Party* (NSDAP), it is worth recalling that the Nazis did not come to power through a coup, but through regular elections. Explaining the success of ERW parties is, therefore, clearly an important issue.

While ERW parties are more heterogeneous than other party families, they share a number of ideological features (Mudde, 1996). In particular, they all have fierce anti-immigration programs, which often become their main focus. Thus, immigration is a natural candidate for explaining the success of ERW parties. Indeed, Figure 1 suggests a positive relationship between the share of immigrants in a population and the support for ERW parties. Taking country fixed effects into account, the correlation between the immigrant share and the ERW vote share is 0.48. (When considering only countries where ERW parties do, in fact, exist, the correlation is 0.51.)

[Insert Figure 1 here]

This correlation is suggestive. However researchers and policy makers are particularly interested in understanding whether immigration in fact *causes* ERW voting. While a large literature has studied the impact of immigration on labor-market outcomes, surprisingly little work has been done to investigate the possible causal relationship between immigration and election outcomes. This paper contributes to closing this gap by investigating the role, in Austria, of immigration as a possible driving force behind the success of the *Freedom Party of Austria* (*Freiheitliche Partei Österreichs*, FPÖ). Until the early 1980s, the FPÖ was a small party with a vote share (in elections to the national parliament) of around 5 percent. When Jörg Haider became the party leader in 1986, the nationalists within the party prevailed over its business-friendly, libertarian wing. After this change, the FPÖ steadily increased its vote share; the nationalistic approach has characterized the party's platform ever since. In 1999, the FPÖ became the country's second-largest party, with a vote share of roughly 27 percent. In 2000, the FPÖ joined with the conservative *Austrian People's Party* (ÖVP) to form a coalition government that was in power until 2006.¹ As suggested by Figure 1 for ERW parties in other European countries, support for the FPÖ seems closely related to its hostility towards immigration.

In this paper, we test the hypothesis that voters in Austrian communities with a higher share of immigrants (residents without Austrian citizenship) are more likely to vote for the FPÖ. The hypothesis is that voting behavior based on anti-immigrant sentiments is influenced by the close geographic proximity of immigrants and natives.

To test this hypothesis, we employ complete, detailed census data, allowing us to sidestep problems of measurement error. Our analysis takes into account the fact that the immigrant share in a given community is not exogenous. The decision of an immigrant as to where to settle is likely to be influenced by the extent of cultural or racial prejudices in a community. Immigrants would rarely settle in communities with a high degree of overt anti-immigrant sentiments, since it would be more difficult to find housing and a job there. Thus, unobserved xenophobia is an omitted variable, leading to a downward bias in a naïve estimation of the effect of the immigrant share on FPÖ vote shares.

¹Internal problems in the FPÖ arose soon after they had become a governing party. These disputes escalated at a special party convention, at which three members of the government resigned. As a result of that a new splinter party, the *Alliance for the Future of Austria*, was established in 2005. Our empirical analysis concerns elections before that date.

Fortunately, the recent history of immigration into Austria offers features that allow us to identify econometrically the effect of the local presence of immigrants on election outcomes. We use historical settlement patterns of immigrants (prior to 1971) as an instrumental variable for the geographic distribution of the immigrant population in later years.² This approach is based on the notion that (i) existing social networks are important elements in the settlement choices of current immigrants, and that (ii) the determinants of the historical settlement patterns are arguably uncorrelated with recent (unobserved) factors of voting behavior. We find strong evidence supporting condition (i). Assumption (ii), the identifying assumption, is not testable. However, we argue that historical settlement patterns form a valid instrument in the Austrian case. Settlement patterns prior to 1971 were not driven by anti-immigrant sentiments. In fact, immigrants arriving in the 1960s were greeted enthusiastically because they provided much-needed labor for the rebuilding of Austria's economy after World War II and during the economic boom of the 1960s.

We also consider the possibility that the native population may change residence in response to a high influx of foreigners. We employ various approaches (as suggested by Peri and Sparber, 2011) to address this issue. It turns out that residential relocations by Austrian voters in response to immigration are not a statistically significant phenomenon. This suggests that our results are not biased by any changes in the composition of the voting populations of communities as a result of immigration.³

In sum, using historical settlement patterns as an instrument for the geographic distribution of contemporaneous immigration seems to be a useful identification strategy in the Austrian

²Other papers have used related identification strategies to investigate the economic effects of immigration (see, for instance, Altonji and Card, 1991; Edin, Fredriksson, and Åslund, 2003; Dustmann, Fabbri, and Preston, 2005; Saiz, 2007; Cortes, 2008). We are not aware of any paper that considers this instrumental variable strategy in the context of political outcomes.

³Contrary to the policies of other countries (such as the U.S.), being born in Austria does not automatically confer citizenship; instead, a child born in Austria must have at least one parent who is an Austrian citizen in order to be entitled to citizenship. While naturalizations are a potential confounding influence on our inferences, they are ultimately unlikely to be important for our results, as we explain further below.

context.

We document two main results. *First*, as hypothesized, the presence of immigrants in their neighborhoods has a quantitatively important and statistically significant impact on citizens' voting patterns; our baseline 2SLS-estimate suggests that a one-percentage-point increase in the share of immigrants in a community increases the percentage of FPÖ votes in general elections by about 0.4 percentage points. This implies that a one-standard-deviation increase in the share of immigrants leads to a quarter of a one-standard-deviation increase in the FPÖ vote share. This effect is larger than the effect implied by the OLS estimates, confirming the importance of controlling for the endogeneity of settlement decisions. We also find that the *increase* in the share of immigrants had a positive effect on the *increase* in the vote share of the FPÖ. The increase in immigration helps to explain an important part of the rise over time in the support for the extreme right.

Our *second* main result shows that the skill composition of immigrants affects voting decisions. We find that the proximity of low- and medium-skilled immigrants causes Austrian voters to turn to the far right. By contrast, high-skilled immigration either has an insignificant or a negative effect on FPÖ votes. This is consistent with the hypothesis that voters vote in their economic interest: High-skilled immigrants improve living conditions for the native population; lower-skilled immigrants pose the greatest threat of labor-market competition. This result is also consistent with the idea that Austrians worry about adverse effects of immigration on the compositional amenities that natives derive from their neighborhoods, schools, and workplaces (Card, Dustmann, and Preston, 2012). Such effects can be expected to play a larger role with low- and medium-skilled immigration than with high-skilled immigration.

Four guideposts can be used to put this analysis into the context of the existing literature. *First*, a significant amount of research and public discussion considers the implications of immigration for the receiving economy in terms of employment, wages, prices, public finances, or racial and cultural features of a society.⁴ However, so far, little evidence exists regarding the causal effects of immigration on election outcomes.⁵

Second, our analysis complements the rich literature, typically based on survey data, on political preferences and attitudes towards immigration. For example, in a recent paper, Dahlberg, Edmark, and Lundqvist (2012) document that immigration reduces preferences for redistribution in Sweden.⁶ The obvious advantage of surveys is that researchers can directly ask the questions they are interested in. For Austria, too, these surveys yield interesting results. For example, analyzing data from the *European and World Values Survey*, we find that those who prefer that scarce jobs be given to native citizens or who even want a complete halt to labor immigration are more likely to be in favor of the FPÖ, as are those who do not care about the living conditions of immigrants or are not willing to do something to improve these conditions. However, surveys also present some problems, sometimes making it difficult to interpret results. In particular, surveys are not anonymous, and survey respondents are unlikely to answer completely truthfully.⁷

Third, our work is related to the literature that studies the political economy of immigration policies. Even in countries where so far no important ERW parties have emerged, immigra-

⁴Indeed, there are now so many reviews of the pertinent literature that it is difficult even to cite all survey papers. An incomplete list of survey articles includes Borjas (1994), Card (2005), Dustmann, Glitz, and Frattini (2008), and Friedberg and Hunt (1995). Longhi, Nijkamp, and Poot (2005) offer a meta-analysis.

⁵Several studies in the political science literature provide suggestive evidence; see, for instance, Arzheimer and Carter (2006); Arzheimer (2009); Golder (2003); Jackman and Volper (1996); Knigge (1998) and Lubbers, Gijsberts, and Scheepers (2002). This literature concludes that high levels of immigration (as well as of unemployment) are positively related to support for ERW parties. However, these empirical findings do not address the endogeneity of immigration and are therefore not able to establish a causal link between immigration and political outcomes. The only exception we are aware of is a study by Gerdes and Wadensjö (2008), examining potential causal effects of asylum seekers from outside Europe and the OECD on voting in Denmark.

⁶For studies on attitudes towards immigration see Card, Dustmann, and Preston (2012); Dustmann and Preston (2004, 2007); Facchini and Mayda (2009); Hainmueller and Hiscox (2007, 2010); Krishnakumar and Müller (2012); O'Rourke and Sinnott (2006); Scheve and Slaughter (2001). For studies related to preferences for political parties, see Citrin, Green, Muste, and Wong (1997); Dülmer and Klein (2005); Knigge (1998); Lubbers and Scheepers (2000).

⁷For example, according to the *European and World Values Survey*, done shortly before the 1999 general election, the FPÖ could expect to obtain about 20 percent of votes, whereas, in the election, the FPÖ scored about 27 percent. Results based on survey data are summarized in the Supplementary Appendix C.

tion policies have been strongly shaped by politico-economic considerations (see, for example, Facchini, Mayda, and Mishra (2011); Facchini and Steinhardt (2011)). Immigration is an issue with a particularly thin line separating pragmatic economic policy from dogmatic political economics. Anti-immigrant politics may have ideological sources, but politicians may also supply xenophobia because they find it instrumental in discrediting political opponents whose policies benefit immigrants (Glaeser, 2005).

Fourth, this paper adds to more general work showing that economic considerations can help explain voting patterns which otherwise seem extreme. Much as economic concerns led many voters to turn to the Nazis (King, Rosen, Tanner, and Wagner, 2008), so have overall economic conditions played a role in the rise of extreme parties in many countries at the beginning of the 20th century (de Bromhead, Eichengreen, and O'Rourke, 2012).

The remainder of this paper is organized as follows. Section 2 discusses the empirical strategy. Section 3 describes the data. Section 4 presents our findings. Section 5 concludes the paper.

2. Empirical model and identification

Our basic approach relates the share of immigrants to the percentage of votes that the FPO obtained in national parliamentary elections. The observation unit is the community, indexed by $i.^8$ We denote by $FP\ddot{O}_{it}$ the percentage of FPÖ votes in community i in election year t; and by IMM_{it} the percentage of immigrants in the resident population in community i at

⁸In Austria, a community is part of a political district, which is in turn part of one of the nine federal states. The community is the lowest administrative level. In 2001, Austria encompassed 2,359 communities in 99 political districts. Vienna is the largest community, with about 1.5 million inhabitants in 2001. For our empirical analysis we divide Vienna into its 23 so-called municipal districts and treat these as separate communities. The smallest community, with 60 inhabitants (in 2001), is Gramais in the federal state of Tyrol. The average community (excluding Vienna) had about 2,800 inhabitants. The number of communities and their territorial boundaries have changed over our sample period. In order to have a balanced panel of communities, we use a slightly modified version of the territorial boundaries of the year 2001, which leaves us with 2,352 communities (including the 23 municipal districts of Vienna).

that time t. Importantly, a simple OLS regression of FPO_{it} on IMM_{it} suffers from potential endogeneity of IMM_{it} . In particular, immigrants are unlikely to be randomly assigned to communities. Instead, immigrants may self-select into communities with low anti-immigration sentiments where jobs and housing are easier to obtain and neighbors are friendlier. If voters with anti-immigrant sentiments are more likely to vote for the FPÖ, ignoring endogeneity of the immigrant share leads to a downward bias of the estimated immigration effect on ERW voting.

To identify the causal effect of immigration on voting outcomes, we need to compare the voting behavior of Austrian citizens in community i after immigration with the counterfactual outcome that would have been observed had immigration not taken place. In observational data, the causal effect can be identified using an instrumental variable, that is, a variable that significantly affects current immigrant shares, while being unrelated to voting decisions except through its effect on immigrant shares.

2.1 Historical settlement patterns as an instrument for the contemporaneous immigrant presence

Our identification strategy relies on historical settlement patterns prior to 1971 as an instrument for immigrant shares in later years. We then use variation in current immigrant shares generated by variation in historical settlement patterns to identify the causal impact of immigration on ERW voting. The key identifying assumption is that historical settlement patterns are not correlated with current unobserved factors determining voting behavior. In other words, historical settlement affects voting today only through the location choice of immigrants arriving later. Thus, historical settlement patterns can be excluded from the second-stage regression. While this identifying assumption is by definition not testable, we argue that this assumption is highly plausible given the historical context of migration to Austria. The argument that historical settlement patterns provide a useful instrument for currently observed regional patterns of immigration was originally proposed by Altonji and Card (1991). Similar instrumental variable strategies have been applied by scholars studying the effects of immigration on wages (for example, Card, 2001) and on prices (for example, Cortes, 2008). However, to our knowledge, this empirical strategy has not been used to identify the impact of immigration on ERW voting. The basic idea is that immigrants settle where they find existing social networks and neighbors with the same cultural and linguistic background (Bartel, 1989; Åslund, 2005; Jaeger, 2007). Such networks facilitate the job search and assimilation into the new cultural environment (Munshi, 2003).⁹

Historical settlement of immigrants to Austria. To understand why historical settlement patterns can serve as a valid instrument, it is important to recall the situation of the Austrian economy in the 1950s and 1960s. The post-war boom of the Austrian economy led to a growing demand for labor amid increasing labor shortages. In the 1960s, the Austrian government began to forge bilateral agreements with southern and southeastern European states to recruit temporary workers. A 1964 agreement with Turkey and a 1966 agreement with Yugoslavia attracted Turkish and Yugoslavian "guest workers" into the country. Recruitment offices in those countries were established, and an influx of Turkish and Yugoslavian workers and their families to Austria began. In 1961, residents with Turkish and Yugoslavian citizenship numbered 271 and 4,565, respectively. By 1971, the numbers had risen to 16,423 and 93,337, respectively. In 1961, the overall number of immigrants was 101,986, equal to 1.4 percent of the overall population. By 1974, mainly as a result of the efforts of the Austrian government to attract guest workers, the number of immigrants had risen to 311,689, equal to 4.1 percent of the overall population. During the 1960s and early 1970s, anti-immigration sentiment was weak. In fact, immigrants were very welcome. The Zeitgeist is well captured by the way the

⁹For the importance of networks in general, see Calvó-Armengol and Jackson (2004), Ioannides and Loury (2004), Lazear (1999), and Montgomery (1991).

first foreign workers arriving from Turkey in 1964 were welcomed in Vienna. Turkish workers were received with cheers of approval and enthusiasm from a large gathering in the Viennese train station. A marching band was playing in their honor and officials handed out flowers to them (*Wiener Zeitung*, 2006/12/30).

In short, settlement decisions of immigrants at that time were not influenced by local cultural or racial prejudices. Immigrant labor was funneled into locations where it brought the greatest marginal benefit.

Further immigration waves and the rise of the FPÖ. The clearly very positive image of immigration of the 1960s and early 1970s started to change in the mid-1970s when the first oil shock pushed Austria into a recession. In response to emerging problems in the labor market, the Austrian government enacted the Aliens Employment Act (1975), which regulated immigration and reduced the influx of foreign workers. This resulted in a period of return-migration and a temporarily stagnating immigrant share.

A second wave of immigration began with the economic boom in the late 1980s. The immigration wave of the late 1980s coincided with the rise of the FPÖ. (For the joint evolution of immigration and FPÖ vote shares, see Figure A.2 in the Supplementary Appendix A.) After Jörg Haider took over leadership of the FPÖ in 1986, the party increasingly invoked the "dangers" to the native population of immigration in terms of crime, unemployment, and decay of neighborhoods and schools. This was accentuated by an additional immigrant wave during the Yugoslavian political crisis in 1990 and the war in 1992. In 1993, the FPÖ launched an "Anti-Foreigner Referendum," and 416, 531 Austrian voters (7.35% of the electorate) approved this referendum. Under political pressure of increased anti-immigration sentiments, and partly as a reaction to the FPÖs anti-immigration activities, the Austrian government enacted various new tighter immigration rules during the 1990s.

Austria's entrance into the EU in 1995 opened the borders to immigration from former EU-

15 member states. In 2002, the center-right coalition of the Austrian People's Party and the FPÖ enacted a set of more restrictive immigration laws.¹⁰

The hypothesis underlying our identification strategy is that the geographic distribution of all these more recent waves of immigrants exhibits strong correlations with the geographic distribution of immigrants from the pre-1971 period. Our first-stage regressions test this hypothesis.

Additional considerations regarding the validity of the identification strategy. To further probe the plausibility of the identifying assumption, we consider three additional points.

First, we discuss possible internal migration effects in Section 4.3; we do not find any significant evidence of such effects.

Second, we note that it is, in principle, conceivable that even the historical settlement patterns prior to 1971 have direct effects on voting behavior today, violating the identifying assumption. This would be the case if, already in 1971, voters in communities where more immigrants had arrived turned to the FPÖ and if there exists intergenerational transmission of voting behavior. As for the first part, the arrival of immigrants was, as discussed above, generally greeted with positive sentiments, as it helped the economy grow. There is no evidence that immigration provided a reason to vote for the FPÖ then. As for the second part, we are not aware of a systematic study of the extent to which voting for the FPÖ is persistent across generations. Perhaps the most direct evidence against this idea is that, in fact, several tectonic shifts have taken place in the Austrian political landscape over the last 50 years. This would not be possible if Austrian voters consistently voted as their parents did. To further address potential time-invariant unobserved heterogeneity, we also estimate a model in differences in

¹⁰These laws included requirements that immigrants study German; restrictions on the temporary workers' ability to obtain permanent residence; and, at the same time, a relaxation of procedures for Austrian firms that were hiring high-skilled immigrants of key importance in certain industries. Further rules were put into place to shield Austria's labor market from excessive immigration from the poor, neighboring, new EU member states after the EU expansions of 2004 and 2007.

Section 4.4.

Third, naturalizations may potentially confound our inferences.¹¹ However, they are unlikely to be important for our results. We first note that they imply two countervailing effects. On the one hand, immigrants who receive Austrian citizenship may still be regarded as immigrants by the "original" Austrian population, so that the immigrant share in our data understates the actual perceived immigrant share in a neighborhood. On the other hand, naturalized immigrants are unlikely to vote for the FPÖ. Second, during the 1970s, 1980s, and 1990s, the annual rate of naturalizations was between 0.1% and 0.3% of the native population in most years. Therefore, we do not attempt to account for naturalizations in our analysis.

Overall, historical patterns of immigrant settlement provide, arguably, a powerful instrument for contemporaneous immigration in our analysis of ERW voting in Austria. The geographic allocation of immigrants in the 1960s was mainly driven by labor shortages in the various regions of Austria; self-selection by immigrants at that time is, therefore, unlikely to contaminate our empirical evidence. Selective migration responses by natives are later shown to be insignificant in the present context.

2.2 The empirical model

We conduct a standard 2SLS approach. Our main analysis considers pooled panel regressions. In all regressions below, we weight observations by community population size. Standard errors are clustered on the community level and robust to heteroskedasticity of unknown form. The first-stage regression is

$$IMM_{it} = \alpha_1 + \beta_1 * IMM_{i,1971} + \mathbf{X}'_{it}\Gamma_1 + \theta_{1t} + \varepsilon_{1it}, \tag{1}$$

¹¹Austrian-born children of immigrants do not obtain Austrian citizenship automatically.

where IMM_{it} denotes the percentage of immigrants in community *i* in a given year, X_{it} is a vector of controls, θ_{1t} is a full set of year dummies, and ε_{1it} is a stochastic error term. $IMM_{i,1971}$ is our instrumental variable.

The second-stage regression then is

$$FP\ddot{O}_{it} = \alpha_2 + \beta_2 * \widehat{IMM}_{it} + \mathbf{X}'_{it}\Gamma_2 + \theta_{2t} + \varepsilon_{2it}, \qquad (2)$$

where $FP\ddot{O}_{it}$ is the percentage of FPÖ votes in community *i* in election year *t*; and \widehat{IMM}_{it} is the predicted value of the percentage of immigrants from the first-stage regression (1). Similar to equation (1), θ_{2t} is a set of year fixed effects, and ε_{2it} is the error term. By including year dummies in both stages, we exploit cross-sectional variation across communities to identify the impact of immigration on ERW voting.

The coefficient of interest is β_2 , which captures the effect of the local presence of immigrants (attracted by existing networks established prior to 1971) on ERW voting. Specifically, β_2 measures the percentage-point change in FPÖ votes that is associated with a one-percentagepoint increase in the immigrant share in a community.

3. Data

Disaggregated community-level data on the percentage of FPÖ votes in elections to the national parliament are available from official statistics issued by the *Austrian Federal Ministry of the Interior*. Figure A.1 in the Supplementary Appendix A shows the geographic distribution of the share of votes for the FPÖ for six general elections. With the exception of a very strong base of support for the FPÖ in the state of Carinthia (located in the south of Austria where former party leader Jörg Haider was leading the local government) no other particular geographical patterns (over time) are evident.

Immigrants are residents without Austrian citizenship. Data on the share of immigrants (on

a community level) are available from the decennial censuses since 1971. Data from 1971 provide the instrumental variable. Since we do not have census data for each possible election year, we need to infer the relevant immigrant share (as well as the socio-economic control variables) in those election years that we wish to analyze. To minimize measurement error, the main analysis focuses on elections that took place at most three years from the time of the nearest census, that is, we consider $t = \{1979, 1983, 1990, 1994, 1999, 2002\}$. We relate the election results of 1979 and 1983 to the 1981 census data. (Consequently, the first stages for 1979 and 1983, when estimated separately for each year, are identical because all the explanatory variables are identical.) Similarly, the election results of 1990 and 1994 are related to the 1991 census data, and the election results of 1999 and 2002 to the 2001 census data.¹²

We also investigate the extent to which ERW voting is driven by the skill composition of immigrants. We calculate immigrant shares within education groups based on residents 25 years of age or older. We sort immigrants into two groups, based on their highest attained education level: *(i) low and medium education* (compulsory schooling, completed apprenticeship training or lower secondary school); and *(ii) high education* (higher secondary school or academic degree).

Our main regressions include a parsimonious set of socio-economic control variables: each community's number of inhabitants (and its square), binary indicators for communities in the states of Vienna and Carinthia (traditionally a FPÖ-stronghold), distribution of marital status (share of inhabitants who are single, married, divorced and widowed), and the population's age-sex-distribution (in five-year age groups). Further robustness checks reported below show that our results are not sensitive to the inclusion of additional controls such as educational attainment and labor-market status.

The immigrant share and all socio-economic control variables are calculated from the universe of all individual-level observations from the decennial Austrian censuses (on-site at *Statis*-

¹²The elections of 1986 and 1995 are not included in the main analysis as they are relatively far from the census dates. However, our results also hold for these years.

tics Austria). The completeness of the census data affords the great advantage that we can sidestep problems of measurement error, an important concern in the literature that studies labor-market effects (Dustmann, Fabbri, and Preston, 2005, p. F329).

Descriptive statistics are in Table 1. Notably, substantial cross-sectional variation exists across communities in Austria, both in election outcomes and immigration levels.

[Insert Table 1 here]

4. Empirical findings

In this section we present our results. We proceed in four steps. In Section 4.1, we provide results based on pooled community data, taking all election years together. We look at both the impact of overall immigration on FPÖ votes and whether (and how) the skill composition of immigration affects FPÖ votes. In Section 4.2, we analyze the various election years separately to see whether the relationship between immigration and FPÖ votes is stable over time. In Section 4.3, we study a potentially important caveat that may invalidate our instrumental variables strategy: migration responses by the voting population. In Section 4.4, we report results of an analysis of the impact of increases in immigration on increases in FPÖ shares.

4.1 The impact of immigration on FPÖ votes

First-stage evidence. The first stage of our identification strategy claims that historical settlement patterns are an important predictor of the contemporaneous immigrant share in a community. To shed light on this issue we first provide some descriptive graphical evidence. The geographic distribution of immigrants by census year is depicted in Figure 2. Visual inspection strongly suggests that the share of immigrants in later years is higher in communities that had a higher share of immigrants in the year 1971. This is illustrated in the three (population-

weighted) scatter plots in Figure 3. The correlations between the immigrant share in 1971 and the corresponding shares in 1981, 1991, and 2001 are 0.82, 0.68, and 0.67, respectively.

[Insert Figures 2 and 3 here]

Panel A of Table 2 shows the first-stage regressions, including an indication of the set of control variables. The specification in the first column concerns all immigrants, the main focus of our analysis. (The second and third column deal with immigrants split into groups by educational attainment; we discuss these results in detail further below.) As expected, the first stage shows a highly statistically significant positive effect of the historical settlement pattern on communities' shares of immigrants in later years.

[Insert Table 2 here]

In sum, the strong correlation between initial settlement patterns and more recent immigrant shares establishes the relevance of the instrument and alleviates weak-instrument concerns.

Second-stage results. Table 3 presents the main results of this paper.

[Insert Table 3 here]

The first column shows a pooled OLS regression, suggesting a positive relationship between immigration and the success of the ERW movement.

Our main inference is, however, based on the 2SLS regression, shown in the third column. The high F-statistics on the excluded instrument suggest that our instrument is sufficiently strong.¹³

The central finding is that the immigrant presence is a highly significant determinant of the percentage of FPÖ votes. Notice that the 2SLS estimate is larger than the OLS estimate. This is consistent with the idea that immigrants self-select into communities where anti-immigrant sentiments are less prevalent. Ignoring this selection would lead the researcher to underestimate the causal effect of immigration on ERW voting. Notably, our 2SLS estimates are almost as precise as the OLS estimates, reflecting that the first stage yields a strong prediction of current immigrant shares.

Immigration is not only a statistically significant but also a quantitatively important predictor of FPÖ votes in the cross-section of Austrian communities. The estimates imply that communities with an immigrant share that is one percentage point higher tend to give about 0.4 percentage points more votes to the FPÖ. Thus, a one-standard-deviation increase in the immigrant share drives about a quarter of a one-standard-deviation increase in the ERW vote share. Note that this local average treatment effect refers only to immigrants attracted by existing networks; immigrants who settled in a certain community for other reasons may have

¹³For the one-instrument case we report Wald F-statistics based on the Cragg-Donald statistic and the Kleibergen-Paap rk statistic. The Cragg-Donald F-statistic is a basic reference point in 2SLS-regressions; Stock, Wright, and Yogo (2002) provide critical values for strong instruments (8.96 in the case of one instrument). However, this statistic requires an assumption of i.i.d. errors. In the presence of clustering and heteroskedasticity, the Kleibergen-Paap rk statistic is, therefore, typically considered additionally in practice. No study appears to exist that provides threshold values that the rk statistic should exceed for weak identification not to be considered a problem, but researchers usually use a value of 10 as an indication of a strong instrument in this case, following the proposal of Staiger and Stock (1997). In the case of multiple endogenous variables, as in our analysis of the role of skill composition, we report the Angrist-Pischke multivariate F-test of excluded instruments. Again, 10 is a threshold value usually employed in practice. In all cases, the cutoff values do not provide a mechanical rule. On the one hand, there is no absolute security that an instrument whose F-statistic exceeds 10 is, indeed, strong; on the other hand, as Angrist and Pischke (2009) point out, even F-statistics as low as 2.0 "may not be fatal" (p. 215). In our main analysis, presented in Table 3, the Angrist-Pischke and Kleibergen-Paap statistics are between 68 and 339, far above conventional thresholds.

a separate effect on FPÖ votes.¹⁴

In terms of control variables,¹⁵ we find important regional variation in the percentages of FPÖ votes; the FPÖ vote share is higher in Carinthia and lower in Vienna. We also find that the FPÖ vote share is significantly affected by community size, the relationship being U-shaped. Among the communities with a population of up to 89,000, the larger communities tend to vote less for the FPÖ; among the communities beyond this critical population level, the larger communities tend to vote more for the FPÖ. Moreover, we find that, in communities with a comparably high share of prime-age women and men above the age of 65, the FPÖ is more successful. Finally, marital status affects FPÖ votes. Communities with a higher share of single (relative to married) individuals tend to vote more for the FPÖ.

The skill composition of immigration. A natural starting point for understanding voting decisions is the hypothesis that rational and self-interested individuals vote for the party which promises them the greatest utility (Downs, 1957). In particular, theories of economic interest (Lipset, 1963) explain that wage, price, and employment effects would be key to understanding voting behavior. Recent survey evidence suggests that, indeed, numerous economic factors play a role in individuals' attitudes toward immigration (Dustmann and Preston, 2004; Hainmueller and Hiscox, 2007). We focus on two ideas.

First, basic economic theory suggests that immigration hurts those native individuals who supply production factors that are close substitutes for factors supplied by immigrant workers.

¹⁴Adding community fixed effects in the OLS-regression shown in the first column of Table 3 allows us to remove time-invariant unobserved heterogeneity. The highly significant relationship between immigration and voting also holds in such a fixed-effects panel estimation (not shown). However, there may be time-variant unobserved heterogeneity, which would not be captured by fixed effects. Moreover, if immigrant levels in community *i* in a given year (for example, in 1991) are negatively related to vote shares for the FPÖ in past years (for example, in 1983), then a fixed-effects estimate of current vote shares for the FPÖ on current immigrant levels will be positively biased. In the 2SLS regressions, we cannot include community fixed effects because our instrumental variable does not vary over time. This is not a great limitation as we are primarily interested in the cross-sectional relationship between immigration and voting decisions. We return to the time-series dimension when we consider difference regressions in Section 4.4.

¹⁵The full regression is shown in Table B.1 in the Supplementary Appendix B.

In contrast, individuals who supply complementary factors will gain from immigration. ERW parties present anti-immigration platforms. If voters are self-interested, those who lose from immigration should, thus, favor ERW parties in elections. The empirical labor-market impact of immigration is strongly debated; some studies (for example, Borjas, 2003) find strong negative effects on native wages, while others do not find strong effects (for example, Card, 2005, 2009).¹⁶ We note that what matters in voting decisions is the perceived impact.

Second, anti-immigration sentiments based on self-interest are related to school quality and neighborhood quality. Card, Dustmann, and Preston (2012), for instance, find that the natives' assessments of "compositional amenities" that they derive from their neighborhoods, schools, and workplaces are an important source of anti-immigration sentiments.

To shed light on this issue, we investigate how the educational levels of immigrants affect voting decisions of natives. Our conjecture is that the natives' voting decisions depend on the skill levels of immigrants either because the intensity of competition for jobs varies across skill levels or because the native population perceives adverse effects on compositional amenities when the immigrants are primarily low- and medium-skilled. We construct two groups of immigrants according to educational attainment, distinguishing between low- and mediumeducation immigrants on the one hand and high-education immigrants on the other hand.

The OLS regression in the second column of Table 3 suggests that low- and mediumeducation immigrants are associated with higher support for the FPÖ, whereas highly educated immigrants are associated with lower support. Here, too, we apply our instrumental variables strategy to get closer to an estimate of the causal effect. We now have two endogenous variables, which are jointly instrumented by the shares of low/medium- and high-education immigrants in the year 1971. As can be seen in the first-stage regressions, in columns two and three of Panel A in Table 2, immigrant networks also work powerfully along the skill dimension. In the later

¹⁶The impact of immigration on the size of the consumer base plays a critical role, complicating theoretical predictions of labor-market effects (Borjas, 2009).

census years, the communities tended to attract and house immigrants of the same educational level as they had in 1971.

Second-stage results show that the same pattern as in the OLS results also holds in the 2SLS setting (fourth column of Table 3). It is the proximity of low- and medium-skilled immigrants which influenced Austrian voters to lean more to the far right.¹⁷

These results provide evidence for anti-immigration sentiments that derive from threats that immigration poses to the labor-market success of natives. High-skilled immigrants bring benefits for the average voter, and they compete for jobs mostly with voters who understand the benefits of the mobility of labor across boundaries and who, therefore, do not find the anti-immigrant position of the FPÖ attractive.

The evidence is also consistent with an explanation based on compositional amenities. In communities where immigration is high-skilled, adverse effects on compositional amenities for the native population are unlikely. In contrast, when immigration is predominantly low- and medium-skilled, anti-immigration sentiments may become stronger as natives perceive an undesired composition of their neighborhoods.

In an attempt to distinguish between the explanations based on labor-market competition and compositional amenities, we checked whether immigrants from different skill-groups have heterogenous effects on FPÖ votes depending on the local share of Austrians with low/medium and high skills. For instance, we run 2SLS regressions including the share of high-skilled Austrians (instrumented by its value in 1971) as well as interaction terms of this share with the shares of low/medium- and high-skilled immigrants (each instrumented by corresponding historical interaction terms). In these regressions, the interaction terms with low- and mediumskilled immigrants are insignificant. This tends to refute an explanation based on labor-market competition: If low-skilled and medium-skilled Austrians worried more about equally-skilled

¹⁷The difference between the 2SLS and the OLS estimates suggests that self-selection into tolerant communities is a particular concern for low- and medium-skilled immigrants.

immigration, as is the case under the labor-market explanation, their votes for the FPÖ would react more than those of high-skilled Austrians; that is, we would expect significantly negative coefficients on the interaction terms. However, this expanded specification is highly demanding (we now have five endogenous variables and five instruments). Therefore, we do not put much emphasis on these specific results concerning skill composition and note that future research is needed to distinguish more explicitly why we observe such strong differences among the effects of different skill groups of immigrants.

Robustness to inclusion of control variables. We check whether the above estimates are sensitive to the inclusion of additional controls. Our basic model presented above used a parsimonious specification (with community characteristics: the number of residents and its square, and a dummy of Carinthia and Vienna; the age-sex distribution of the resident population; the distribution of marital status among residents; and election-year fixed effects). We chose this lean specification because many characteristics of the resident population may themselves be influenced by immigration (for instance, via their participation in the local labor market), constituting, therefore, potentially "bad controls" (Angrist and Pischke, 2009). Table B.2 in the Supplementary Appendix B shows that the estimated 2SLS effects of immigration on FPÖ votes vary only very little across specifications where we (i) exclude all control variables, (ii) add the distribution of educational attainment, or (iii) include in addition the distribution of labor-market status.¹⁸ These specifications also reveal that communities with a higher share of medium- and low-skilled residents tend to lean more towards the FPÖ.

Robustness to functional form. We also consider several different functional forms to model the impact of immigration on FPÖ votes. For example, we add a quadratic term of the immigration share to our model. Alternatively, we try a flexible specification based on binary

¹⁸The Austrian Census does not collect information on income. However, information on educational attainment and labor-market status should proxy well for income.

variables capturing quartiles of the share of immigrants. While the (adapted) first stage is again very strong in each case, we do not find economically relevant, systematic non-linearities in the second-stage estimation. We conclude that the simple linear model captures the immigration effect quite well.

4.2 Regressions by election years

Has the relationship between immigration and FPÖ votes changed over time, or has it been stable? We consider separate regressions for each election year. These regressions use the same community, family status, and age- and sex-distribution controls as the pooled regressions. The second-stage results are summarized in Panel A of Table 4.¹⁹

[Insert Table 4 here]

In each election year we find a significant positive effect of the share of immigrants in a community on the share of votes for the FPÖ. Comparing the estimated effects with those from the OLS regressions, shown in Panel B of Table 4, we can again see that the OLS estimates tend to be downward biased.²⁰

The size of the estimated effect of immigration on the share of votes for the FPÖ varies only little across election years. The highest effect occurred in 1979. In most years since then, the FPÖ has received a vote share that is approximately 0.3 percentage points higher per additional percentage point of immigrant share. In sum, the strength of the relationship between immigration and FPÖ votes seems rather stable over time and does not follow any

¹⁹The first stages (not shown in Table 4) remain strong. Note that the first-stage regressions for election year pairs $\{1979, 1983\}$, $\{1990, 1994\}$, and $\{1999, 2002\}$ are identical because we match election year data to the census closest to the respective election years.

 $^{^{20}}$ We obtain similar results for those election years which were not considered in the main analysis because of their distance from the nearest census.

particular trend.²¹

When we perform the analysis that accounts for the skill composition of immigration, this basic conclusion is also confirmed. The second-stage findings are summarized in Table 5. The overall pattern in these results is quite similar to that of the pooled panel regression. In all years, low- and medium-skilled immigration had a significantly positive effect on Austrians' decisions to vote for the FPÖ. For high-skilled immigration we find in the first year, 1979, do the estimations suggest (albeit insignificantly) that voters may have seen high-skilled immigration as a reason to turn to the FPÖ, whereas in later years more high-skilled immigration did not benefit (and in fact hurt) the ERW movement.

[Insert Table 5 here]

Overall, this analysis shows that the general picture is quite robust and that the same pattern of results that we observed for the pooled sample shows up also in the year-by-year analysis.

4.3 Internal migration of voters

Austrian voters are free in their residential location choices within the country (and the EU). Hence, our results may be contaminated by internal migration responses of Austrians. To the extent that such voter relocations are important, our results are likely to *underestimate* the true effect of immigration on FPÖ voting. This is because the voters whose welfare is negatively affected by the proximity of immigrants (and who would, therefore, more readily gravitate to the FPÖ) are more likely to have moved elsewhere.

To test for the importance of native internal migration responses, we follow Peri and Sparber

 $^{^{21}}$ It is difficult to detect a systematic pattern that could plausibly explain the variation across election years. There seems to be no systematic relationship between the size of the estimated effect and the following possible explanations: (i) the overall share of votes for the FPÖ, (ii) the FPÖ's top candidate, (iii) the major topics in the election campaigns, (iv) any business cycle indicator, or (v) the absolute time lag between the election data and the census year which might give rise to an attenuation bias.

(2011). The question is how many natives (N) respond to the arrival of immigrants (I) by leaving their place of residence *i*. To estimate the quantitative importance of such migration responses, the following model is estimated: $\Delta N_{i,t} = \alpha + \beta \cdot \Delta I_{i,t} + u_{i,t}$ with β being the interesting parameter. Various scholars have proposed different versions of this model, mainly considering different measurement concepts of dependent and independent variables.

Table 6 summarizes the estimation output of three empirical models for our community-level panel data, with *i* communities over *t* years, where $i = \{1, ..., 2352\}$ and $t = \{1971, 1981, 1991, 2001\}$. Since we are concerned with the whole population (and not only with the labor force), our sample is based on community-year cells and abstracts from the skill dimension. Specification (1), a slightly modified specification of Card (2001, 2007), is the preferred specification of Peri and Sparber (2011). This specification provides no evidence for any internal migration response of Austrians. Even based on specifications (2) and (3)–which Peri and Sparber (2011) verify to be biased towards an attraction and a displacement effect, respectively–we do not find any statistically significant effect. This evidence is in line with the common stereotype that the Austrian population is very rooted. Overall, these findings support the validity of our identification strategy.

[Insert Table 6 here]

4.4 Estimates based on first differences

Our above analysis aims at explaining the cross-sectional variance across communities to test the hypothesis that immigration increased votes for the extreme right in Austria.

A different, but related, question is whether the rise of the FPÖ can be explained by the increase in immigration. In 1961, only 1.4 percent of the resident Austrian population were foreign citizens; by 1981, this share had almost tripled. From 1981 to 2001, the share of immigrants more than doubled again, from 3.9 to 8.7 percent, with much variation across

communities. The development over time of the share of votes for the FPÖ is strikingly similar. Until 1986, the FPÖ had not played a significant role in national elections (despite having been a junior partner in a government coalition). In the national elections of 1986, however, the FPÖ attracted 9.7 percent of the votes. Thereafter, support for the FPÖ grew at a steady rate, passing the 15 percent and 20 percent hurdles in 1990 and 1994, respectively, and reaching more than 25 in the late 1990s. Figure A.2 shows these two developments.

Making sense of the aggregate correlation is difficult because the observed correlation may be due to other events of that time.²² We can make progress by analyzing the consequences of the increased presence of immigrants in any particular community. The question is whether the *rise* in FPÖ votes is concentrated in communities that experienced a disproportionate *increase* in immigration. In other words, rather than exploiting the cross-sectional variation in *levels* of FPÖ votes and immigrant shares, we exploit the cross-sectional variation in *changes* in FPÖ votes and immigrant shares. Just as in our basic model above, we rely on settlement patterns in 1971 to instrument the increase in immigration since that year. Generally, the first stages in this analysis also perform well, although they are somewhat less strong than in the levels-based regressions.

[Insert Table 7 here]

The estimation results for overall immigrant shares are summarized in Panel A of Table 7. The first column shows the (second-stage) results from the pooled panel, whereas the remaining columns show the results by election year. The 2SLS estimate is mostly significant and quite large, and the implied quantitative effects are substantial. For example, a one-percentage-point increase in immigration from 1971 to 1999 generates 1.41 percentage points of additional FPÖ votes in 1999, compared to 1971. The increase in the immigrant share in that time period was about 6 percentage points, and the increase in the FPÖ vote share was about 21.5 percentage

 $^{^{22}}$ For example, the Austrian political landscape in the 1990s was also characterized by a general dissatisfaction with the governing parties. The *Social Democratic Party of Austria* and the *Austrian People's Party* had been governing as a grand coalition since 1987.

points. Thus, about a two-fifths (= $6 \times 1.41/21.5$) of the total rise of the FPÖ in this time span can be explained by immigration.²³ In sum, the quantitative implications that are obtained from exploiting cross-community variation in increases of immigrant shares and FPÖ vote shares are similar to the picture we get from exploiting cross-community variation in levels of immigrant shares and FPÖ vote shares.

Panel B of Table 7 summarizes the estimation results by the skill composition of immigrants. In the pooled sample, we find a positive effect of an increase in low- and medium-education immigrants on an increase in FPÖ vote shares, and no significant effect of high-education immigrants. The positive effect of low- and medium-education immigrants is also present (and mostly statistically significant) in the estimations by election year. Although the point estimates on the increase in high-education immigrants is large, this effect is less precisely estimated in these specifications, suggesting overall no significant effect of the increase of the share of this group of immigrants on the increase in FPÖ vote shares. These results are again consistent with the previous analysis based on levels of immigrant shares and FPÖ vote shares.

5. Conclusions

Political folklore holds that ERW parties attract voters by appealing to anti-immigration sentiments of the voting population. While existing empirical studies in the (predominantly political science) literature provide support for a correlation between immigration and votes for the extreme right, the causal impact of immigration on voting for the extreme right has not yet been established.

This paper contributes to closing this gap. Studying the rise of the right-wing Freedom Party of Austria (FPÖ) that has occurred since the mid-1980s, we establish two main results.

 $^{^{23}}$ The cross-sectional standard deviations of the increases in immigrant shares and FPÖ vote shares, respectively, were 5.2% and 4.9%. Thus, cross-sectional variation in increases implies essentially a one-to-one variation in FPÖ vote shares. Virtually the same results hold when controlling for the initial level of the FPÖ vote share.

First, we find that roughly a quarter of the cross-community variation in the percentage of FPÖ votes can be attributed to cross-community variation in the presence of immigrants. We also find that the increase in the local share of immigrants had a positive effect on the increase in the local vote share of the FPÖ. *Second*, the skill composition of immigrants affects voting decisions. It is the proximity of low- and medium-skilled immigrants that causes Austrian voters to turn to the far right. High-skilled immigration either has an insignificant or a negative effect on FPÖ votes.

We obtain these results using an instrumental variables strategy. Specifically, past settlement patterns of immigrants in Austrian communities have great predictive power for the more recent cross-community variation of immigrant shares. Because the historical settlement pattern is unlikely to be related to voting behavior today, it can serve as an instrument for the local presence of immigration in recent election years, allowing identification of the causal effect of local immigration on local FPÖ votes. Further support for the validity of our identification strategy comes from the finding that immigration did not significantly affect voters' residential choices.

It is a simple fact that immigration is necessary for modern economies, given demographic developments such as a persistently low fertility rate and the ageing of society. However, immigration also has potentially critical political implications, including the possible rise of extreme-right-wing parties. Several channels are likely to exist through which immigration may affect voting decisions, and each channel requires different policy responses. What our paper shows is that the geographic proximity of immigrants is one economically and statistically significant causal driver behind the support for the far right. This result has important policy implications. In particular, the evidence suggests that policies at the local level deserve significant attention. For example, it is possible that integration policies in the community can help restrict emerging xenophobia. Future work is needed to understand which specific policies are particularly suitable.

References

ALMOND, G., AND S. VERBA (1965): The Civic Culture. Little, Brown, Boston.

- ALTONJI, J., AND D. CARD (1991): "The Effects of Immigration on the Labor Market Outcomes of Natives," in *Immigration, Trade, and the Labour Market*, ed. by J. M. Abowd, and R. B. Freeman, pp. 201–234. University of Chicago Press for NBER, Chicago.
- ANGRIST, J. D., AND J. PISCHKE (2009): Mostly Harmless Econometrics: An Empiricist's Companion. Princeton University Press, Princeton, NJ.
- ARZHEIMER, K. (2009): "Contextual Factors and the Extreme Right Vote in Western Europe, 1980–2002," American Journal of Political Science, 53(2), 259–275.
- ARZHEIMER, K., AND E. CARTER (2006): "Political Opportunity Structures and Right-Wing Extremist Party Success," *European Journal of Political Research*, 45, 419–443.
- BARTEL, A. P. (1989): "Where Do the New United States Immigrants Live?," Journal of Labor Economics, 7(4), 371–391.
- BORJAS, G. J. (1994): "The Economics of Immigration," *Journal of Economic Literature*, 32(4), 1667–1717.
- (2003): "The Labor Demand Curve Is Downward Sloping: Reexamining the Impact of Immigration on the Labor Market," *Quarterly Journal of Economics*, 118(4), 1335–1374.
- (2009): "The Analytics of the Wage Effect of Immigration," NBER Working Paper 14796, National Bureau of Economic Research, Cambridge, MA.
- CALVÓ-ARMENGOL, A., AND M. O. JACKSON (2004): "The Effects of Social Networks on Employment and Inequality," *American Economic Review*, 94(3), 426–454.

- CARD, D. (2001): "Immigrant Inflows, Native Outflows, and the Local Labor Market Impacts of Higher Immigration," *Journal of Labor Economics*, 19(1), 22–64.
- (2005): "Is the New Immigration Really So Bad?," *Economic Journal*, 115, F300–F323.
- (2007): "How Immigration Affects US Cities," CReAM Discussion Paper 11, Centre for Research and Analysis of Migration, London.
- (2009): "Richard T. Ely Lecture: Immigration and Inequality," *American Economic Review*, 99(2), 1–21.
- CARD, D., C. DUSTMANN, AND I. PRESTON (2012): "Immigration, Wages, and Compositional Amenities," *Journal of the European Economic Association*, 10(1), 78–119.
- CITRIN, J., D. P. GREEN, C. MUSTE, AND C. WONG (1997): "Public Opinion Toward Immigration Reform: The Role of Economic Motivations," *Journal of Politics*, 59(3), 858– 881.
- CORTES, P. (2008): "The Effect of Low-skilled Immigration on US Prices: Evidence from CPI Data," *Journal of Political Economy*, 116(3), 381–422.
- DAHL, R. A. (1989): Democracy and Its Critics. Yale University Press, New Haven.
- DAHLBERG, M., K. EDMARK, AND H. LUNDQVIST (2012): "Ethnic Diversity and Preferences for Redistribution," *Journal of Political Economy*, 120(1), 41–76.
- DE BROMHEAD, A., B. EICHENGREEN, AND K. H. O'ROURKE (2012): "Right-wing Political Extremism in the Great Depression," NBER Working Paper 17871, National Bureau of Economic Research, Cambridge, MA.
- DOWNS, A. (1957): An Economic Theory of Democracy. Harper and Row, New York.

- DÜLMER, H., AND M. KLEIN (2005): "Extreme Right-wing Voting in Germany in a Multilevel Perspective: A Rejoinder to Lubbers and Scheepers," *European Journal of Political Research*, 44, 243–263.
- DUSTMANN, C., F. FABBRI, AND I. PRESTON (2005): "The Impact of Immigration on the British Labour Market," *Economic Journal*, 115, F324–341.
- DUSTMANN, C., A. GLITZ, AND T. FRATTINI (2008): "The Labour Market Impact of Immigration," Oxford Review of Economic Policy, 24(3), 477–494.
- DUSTMANN, C., AND I. P. PRESTON (2004): "Is Immigration Good or Bad for the Economy? Analysis of Attitudinal Responses," *Research in Labour Economics*, 24, 3–34.
- (2007): "Racial and Economic Factors in Attitudes to Immigration," *The B.E. Journal* of *Economic Analysis & Policy: Advances*, 7(1), Article 62.
- EDIN, P., P. FREDRIKSSON, AND O. ÅSLUND (2003): "Ethnic Enclaves and the Economic Success of Immigrants: Evidence from a Natural Experiment," *Quarterly Journal of Economics*, 119, 329–357.
- FACCHINI, G., AND A. MAYDA (2009): "Does the Welfare State Affect Individual Attitudes Toward Immigrants? Evidence Across Countries," *Review of Economics and Statistics*, 91(2), 295–314.
- FACCHINI, G., A. MAYDA, AND P. MISHRA (2011): "Do Interest Groups Affect US Immigration Policy?," Journal of International Economics, 85(1), 114–128.
- FACCHINI, G., AND M. STEINHARDT (2011): "What Drives US Immigration Policy? Evidence from Congressional Roll Call Votes," *Journal of Public Economics*, 95, 734–743.
- FRIEDBERG, R. M., AND J. HUNT (1995): "The Impact of Immigrants on Host Country Wages, Employment and Growth," *Journal of Economic Perspectives*, 9(2), 23–44.

- GERDES, C., AND E. WADENSJÖ (2008): "The Impact of Immigration on Election Outcomes in Danish Municipalities," IZA Discussion Paper 3586, Institute for the Study of Labor, Bonn, Germany.
- GLAESER, E. L. (2005): "The Political Economy of Hatred," *Quarterly Journal of Economics*, 120(1), 45–86.
- GOLDER, M. (2003): "Explaining Variation in the Success of Extreme Right-Wing Parties in Western Europe," Comparative Political Studies, 36(4), 432–466.
- HAINMUELLER, J., AND M. J. HISCOX (2007): "Educated Preferences: Explaining Attitudes Toward Immigration in Europe," *International Organization*, 61(2), 399–442.
- (2010): "Attitudes toward Highly Skilled and Low Skilled Immigration: Evidence from a Survey Experiment," *American Political Science Review*, 104(1), 61–84.
- IOANNIDES, Y. M., AND L. D. LOURY (2004): "Job Information Networks, Neighborhood Effects, and Inequality," *Journal of Economic Literature*, 42(4), 1056–1093.
- JACKMAN, R. W., AND K. VOLPER (1996): "Conditions Favouring Parties of the Extreme Right in Western Europe," *British Journal of Political Science*, 26(4), 501–521.
- JAEGER, D. A. (2007): "Green Cards and the Location Choices of Immigrants in the United States, 1971-2000," Research in Labor Economics, 27, 131–183.
- KING, G., O. ROSEN, M. TANNER, AND A. F. WAGNER (2008): "Ordinary Economic Voting Behavior in the Extraordinary Election of Adolf Hitler," *Journal of Economic History*, 68(4), 951–996.
- KNIGGE, P. (1998): "The Ecological Correlates of Right-Wing Extremism in Western Europe," European Journal of Political Research, 34(2), 249–279.

- KRISHNAKUMAR, J., AND T. MÜLLER (2012): "The Political Economy of Immigration in a Direct Democracy: The Case of Switzerland," *European Economic Review*, 56(2), 174–189.
- LAZEAR, E. P. (1999): "Culture and Language," Journal of Political Economy, 107(S6), S95– S126.
- LIPSET, S. M. (1963): *Political Man: The Social Basis of Politics*. Anchor Books, Garden City, NY.
- LONGHI, S., P. NIJKAMP, AND J. POOT (2005): "A Meta-Analytic Assessment of the Effect of Immigration on Wages," *Journal of Economic Surveys*, 19(3), 451–477.
- LUBBERS, M., M. GIJSBERTS, AND P. SCHEEPERS (2002): "Extreme Right-wing Voting in Western Europe," *European Journal of Political Research*, 41(3), 345–378.
- LUBBERS, M., AND P. SCHEEPERS (2000): "Individual and Contextual Characteristics of the German Extreme Right-Wing Vote in the 1990s. A Test of Complementary Theories," *European Journal of Political Research*, 38(1), 63–94.
- MAYDA, A. (2006): "Who Is Against Immigration? A Cross-Country Investigation of Individual Attitudes toward Immigrants," *Review of Economics and Statistics*, 88(3), 510–530.
- MONTGOMERY, J. D. (1991): "Social Networks and Labor-Market Outcomes: Toward an Economic Analysis," *American Economic Review*, 81(5), 1408–1418.
- MUDDE, C. (1996): "The War of Words. Defining the Extreme Right Party Family," West European Politics, 19(2), 225–248.
- MUNSHI, K. (2003): "Networks in the Modern Economy: Mexican Migrants in the U.S. Labor Market," *Quarterly Journal of Economics*, 118(2), 549–599.

- O'ROURKE, K. H., AND R. SINNOTT (2006): "The Determinants of Individual Attitudes towards Immigration," *European Journal of Political Research*, 22(4), 838–861.
- PERI, G., AND C. SPARBER (2011): "Assessing Inherent Model Bias: An Application to Native Displacement in Response to Immigration," *Journal of Urban Economics*, 69(1), 82–91.
- ASLUND, O. (2005): "Now and Forever? Initial and Subsequent Location Choices of Immigrants," *Regional Science and Urban Economics*, 35(2), 141–165.
- SAIZ, A. (2007): "Immigration and Housing Rents in American Cities," Journal of Urban Economics, 61(2), 345–371.
- SCHEVE, K. F., AND M. J. SLAUGHTER (2001): "Labor Market Competition and Individual Preferences Over Immigration Policy," *Review of Economics and Statistics*, 83(1), 133–145.
- STAIGER, D., AND J. H. STOCK (1997): "Instrumental variables regression with weak instruments," *Econometrica*, 65(3), 557–586.
- STOCK, J. H., J. H. WRIGHT, AND M. YOGO (2002): "A Survey of Weak Instruments and Weak Identification in Generalized Method of Moments," *Journal of Business and Economics Statistics*, 20(4), 518–529.



Figure 1. Immigration and ERW-voting in the EU-15 countries, Norway, and Switzerland, 1970–2008

This scatterplot accounts for country fixed effects (i.e., both variables are centered around the respective country-specific mean) and is based on 119 general election vears in EU-15 countries. Norway, and Switzerland in the period between 1970 and 2008; only democratic periods are used. 65 elections are not included due to missing information on the absolute number of residents without citizenship. Share of immigrants is defined as the absolute number of residents without citizenship relative to all residents. Data on the total number of residents is from the database of *Eurostat*. Information on the absolute number of residents without citizenship is from various national sources; details are available upon request. Share of votes for extreme-right wing (ERW) parties include the following parties: Austria: sum of the following two parties: (i) Freiheitliche Partei Österreichs founded in 1956, (ii) Bündnis Zukunft Österreich founded 2005 as a splinter from the Freiheitliche Partei Österreichs, parliamentary elections (National Council of Austria); Belgium: sum of the following three parties: (i) Vlaams Blok founded in 1978 and succeeded by the Vlaams Belang in 2004, (ii) Le Front national founded in 1985, (iii) Lijst Dedecker founded in 2007, general elections (Belgian Chamber of Representatives); Denmark: sum of the following two parties: (i) Fremskridtspartiet founded in 1972, (ii) Dansk Folkeparti founded 1995 as a splinter from the Fremskridtspartiet, parliamentary elections (Danish Parliament); Finland: Suomen maaseudun puolue founded in 1959, dissolved de facto in 1995 (de jure in 2003), and succeeded by Perussuomaliset founded in 1995, Finnish parliamentary elections; France: Front National founded in 1972, French legislative elections (first round votes); Germany: sum of the following two parties (i) Nationaldemokratische Partei Deutschlands – Die Volksunion founded in 1964, (ii) Die Republikaner founded in 1983, German federal elections; Greece: sum of the following two parties: (i) Ethniki Parataxis founded in the late 1970s, (ii) Laikós Orthódoxos Synagermós founded in 2000, Greek legislative election; Ireland: no ERW-parties; Italy: sum of the following two parties: (i) Movimento Sociale Italiano-Destra Nazionale founded in 1946, dissolved in 1995, and transformed into the Alleanza Nazionale (dissolved 2009), (ii) Lega Nord founded in 1991, Italian general elections; Luxembourg: no ERW-parties; Netherlands: sum of the following three parties: (i) Centrumpartij founded in 1980 and dissolved in 1986, (ii) Lijst Pim Fortuyn founded in 2002 and dissolved in 2008, (iv) Partij voor de Vrijheid founded in 2006, Dutch general elections; Norway: Framstegspartiet founded in 1973, Norwegian parliamentary elections; Portugal: Partido Popular Monárquico founded in 1974, Portuguese legislative elections; Spain: no ERW-parties; Sweden: Sverigedemokraterna founded in 1988, Swedish general elections; Switzerland: Schweizerische Volkspartei founded in 1971, Swiss federal elections; United Kingdom: no ERW-parties. Data on election results are obtained from the Comparative Political Data Set I (23 OECD Countries) provided by Klaus Armingeon, Sarah Engler, Panajotis Potolidis, Marléne Gerber and Philipp Leimgruber (see http://www.ipw.unibe.ch/content/team/klaus_armingeon/comparative_political_data_sets/index_ger.html). Information on founding years is from Wikipedia.



Figure 2. The spatial distribution of immigrants by census year^a

These figures depict the share of immigrants (defined as the number of residents without Austrian citizenship as a percent share of all residents) in Austrian communities in the census years 1971, 1981, 1991, and 2001. The number of communities and their territorial boundaries has changed over the sample period. In order to have a balanced panel of communities, a slightly modified version of the territorial boundaries of the year 2001 with 2,352 communities (including the 23 municipal districts of Vienna) is used.





36

These population-weighted scatter-plots (based on Austrian community-level data from the decennial Austrian census) depict the correlation between the share of immigrants in Austrian communities in the year 1971, and in those in the years 1981, 1991 and 2001. For presentational purposes, the graphs exclude communities with more than 20 percent of immigrants. In case of the first graph there are 3, in the second 9, and in the third 17 of such outliers. These observations are included in the empirical analysis.

Election year	Pooled	1971	1979	1983	1990	1994	1999	2002
% share of FPÖ-votes	14.84	5.49	6.10	5.03	16.68	22.81	27.39	10.23
% share of immigrants	(9.10) 6.50 (5.67)	(3.08) 2.83 (2.56)	(3.72) 3.86 (3.75)	(3.18) 3.86 (3.75)	(5.73) 6.64 (5.42)	(5.43) 6.64 (5.42)	(5.99) 8.85 (6.30)	(4.78) 8.85 (6.30)
with low and medium skills	(5.07) (5.20) (4.92)	(2.30) (2.45)	(3.13) (3.21) (3.47)	(3.13) (3.21) (3.47)	(5.42) 5.27 (4.74)	(5.42) 5.27 (4.74)	(0.50) 7.01 (5,50)	(0.50) 7.01 (5.50)
with high skills	(1.02) 1.14 (1.39)	(2.16) 0.36 (0.55)	(0.11) 0.49 (0.76)	(0.11) 0.49 (0.76)	(1.11) (1.20) (1.37)	(1.11) (1.20) (1.37)	(0.00) 1.68 (1.60)	(0.00) 1.68 (1.60)

 Table 1. Descriptive statistics on variables of primary interest

This table summarizes population-weighted means and standard deviations (in parentheses below) for the variables of primary interest based on Austrian community-level data. The share of votes for the FPÖ is from general elections; these figures might differ slightly from official election results due to overseas voters and varying turnout of voters across communities. The share of immigrants (with a certain level of education) is equal to the number of residents without Austrian citizenship (with the respective educational attainment) as a fraction of all residents. Shares by skill are calculated based on residents 25 years of age or older and refer to the highest attained educational degree. Low and medium education is compulsory schooling, an apprenticeship or a lower secondary school. High education is a higher secondary school or an academic degree. The shares of immigrants on a community-level are only available in the years 1971, 1981, 1991 and 2001 (census years). The shares of immigrants in the years 1979 and 1983 are imputed with information form the year 1981, the data in the years 1990 and 1994 are imputed with information form the years 1999 and 2002 are imputed with information from the year 2001.

	(1) Share of immigrants overall		(2) Share immigrants low-& medium	of with n skills	(3) Share of immigrants with high skills			
Panel A: Pooled sample Share of immigrants in 1971 with low and medium skills with high skills	0.873***	(0.047)	0.803*** 0.280	(0.044) (0.275)	0.034^{***} 0.678^{***}	(0.010) (0.080)		
Community ^{a} Family status ^{b} Age-sex-distribution ^{c} Year fixed effects ^{d}	Yes Yes Yes	Yes Yes Yes Yes Yes Yes Yes Yes		YesYesYesYesYesYesYesYes		Yes Yes Yes Yes		
Panel B1: 1981 sample ^{e} Share of immigrants in 1971 with low and medium skills with high skills	0.910***	(0.033)	0.831^{***} 0.264^{*}	(0.038) (0.141)	0.049^{***} 0.518^{***}	(0.018) (0.073)		
Panel B2: 1991 sample ^e Share of immigrants in 1971 with low and medium skills with high skills	0.878***	(0.054)	0.804*** 0.234	(0.050) (0.262)	0.041^{***} 0.611^{***}	(0.015) (0.086)		
Panel B3: 2001 sample ^{e} Share of immigrants in 1971 with low and medium skills with high skills	0.849***	(0.064)	0.758^{***} 0.246	(0.064) (0.451)	0.057^{***} 0.701^{***}	(0.015) (0.096)		

Table 2. First stage: Determinants of the share of immigrants (by education)

This table summarizes estimations of the determinants of the share of immigrants (i.e. residents without Austrian citizenship), the share of immigrants with low and medium education (compulsory schooling, apprenticeship or lower secondary school), and the share of immigrants with high education (higher secondary school or academic degree) in community *i* in the year *t*, where $i = \{1, \ldots, 2, 352\}$ and $t = \{1981, 1991, 2001\}$ based on Austrian community-level census data. Details on the calculation of the share of immigrants (by educational attainment) are provided in the notes to Table 1. Method of estimation is OLS with community population weights. Robust standard errors (allowing for clustering on a community level and/or and heterosekdasticity of unknown form) are in parentheses. *, ** and *** indicate statistical significance at the 10-percent level, 5-percent level, and 1-percent level, respectively. *a* Community's number of inhabitants, number of inhabitants squared, and binary variables indicating, communities in Vienna, and Carinthia. ^b Distribution of sex *s* and in age-group *a*, where *a* is one of sixteen age groups 0-5, 5-10, ..., 70-75, 80+. ^d Base year: 1981. ^e The first stages for the three individual years 1981, 1991, and 2001 include the same control variables as the pooled sample regression (except year fixed effects).

	(1) OLS	$(2) \\ OLS$		(3) 2SLS	(4) 2SLS	
Immigration Share of immigrants with low and medium skills with high skills	0.227^{***} (0.027)	0.285^{***} -0.286^{***}	(0.029) (0.077)	0.386^{***} (0.045)	0.431^{***} -0.244	(0.054) (0.292)
$\operatorname{Community}^{a}$	Yes	Yes		Yes	Yes	
Family status ^{b}	Yes	Yes		Yes	Yes	
Age-sex-distribution ^{c}	Yes	Yes		Yes	Yes	
Year fixed effects ^{d}	Yes	Yes		Yes	Yes	
<i>F</i> -statistics on excl. instrument(s)	-	_		$6,983/339^{e}$	270/68	f

Table 3. The effect of immigration on FPÖ votes: OLS and 2SLS pooled estimations

This table summarizes the estimated effect of immigration (share of residents without Austrian citizenship) on the share of votes for the FPÖ based on a series of weighted (community population weights) OLS (first two columns) and instrumental variable (third and fourth column) estimations using Austrian community level data. The dependent variable is equal to the share of votes for the FPÖ in the general election in community *i* in the year *t*, where $i = \{1, \ldots, 2, 352\}$ and $t = \{1979, 1983, 1990, 1994, 1999, 2002\}$. In column (3) the endogenous variable is the share of immigrants in community *i* in year *t*, which is instrumented by the share of immigrants in community *i* in year 1971. In column (4) the endogenous variables are the shares of immigrants with low/medium and high education, which are instrumented by the 1971 shares of immigrants with low/medium and high education. Details on the calculation of the share of immigrants by educational attainment are provided in the notes to Table 1. Robust standard errors (allowing for clustering on a community level and/or and heterosekdasticity of unknown form) are in parentheses. *, ** and *** indicate statistical significance at the 10-percent level, 5-percent level, and 1-percent level, respectively. *a* Community's number of inhabitants, number of inhabitants squared, and binary variables indicating, communities in Vienna, and Carinthia. *b* Distribution of marital status: shares of inhabitants who are single, married, divorced or widowed. *c* 34 variables that capture the share of the total population of sex *s* and in age-group *a*, where *a* is one of sixteen age groups 0-5, 5-10, ..., 70-75, 80+. *d* Base year: 1981. *e* Wald *F*-statistics based on the Cragg-Donald statistic/Kleibergen-Paap *rk* statistic. *f* Angrist-Pischke multivariate *F*-test of excluded instruments.

Election year	1979	1983	1990	1994	1999	2002
Panel A: 2SLS regressions 2nd stage: $\beta_2^{2\hat{S}LS}$	0.431^{***} (0.063)	0.230^{***} (0.040)	0.270^{***} (0.065)	0.176^{***} (0.059)	0.338^{***} (0.063)	0.357^{***} (0.042)
$F\mbox{-statistics}$ on excl. $\mbox{instrument}^a$	3,277/743	3,277/743	1,202/263	1,202/263	746/175	746/175
Panel B: OLS regressions $\beta^{\hat{O}LS}$	0.354^{***} (0.059)	0.217^{***} (0.046)	0.244^{***} (0.043)	0.224^{***} (0.039)	0.233^{***} (0.033)	0.191^{***} (0.020)

Table 4. The effect of immigration on FPÖ votes: 2SLS and OLS cross-section estimations

This table summarizes the estimated effect of immigration (share of residents without Austrian citizenship) on the share of votes for the FPÖ based on a series of weighted (community population weights) instrumental variable (panel A) and OLS (panel B) estimations using Austrian community level data. The dependent variable is equal to the share of votes for the FPÖ in the general election in community *i* in the year *t*, where $i = \{1, \ldots, 2, 352\}$ and $t = \{1979, 1983, 1990, 1994, 1999, 2002\}$. The endogenous variable is the share of immigrants in community *i* in that year, which is instrumented in the 2SLS estimations by the share of immigrants in community *i* in the year 1971. The shares of immigrants in community *i* are available in the years 1981, 1991 and 2001. The share of immigrants in the years 1979 and 1983 is imputed with information form the year 1981, the data in the years 1990 and 1994 are imputed with information form the year 1991, and the data in the years 1990 and 2002 are imputed with information form the year 1991, and the data in the years 1990 and 2002 are imputed with information form the year sequence of inhabitants, number of inhabitants squared, the distribution of marital status (shares of inhabitants who are single, married, divorced or widowed), the whole age-sex-distribution (34 groups), and binary variables indicating, communities in Vienna, and Carinthia. Robust standard errors (allowing for heterosekdasticity of unknown form) are in parentheses. *, *** and *** indicate statistical significance at the 10-percent level, 5-percent level, and 1-percent level, respectively. *a* Wald *F*-statistics based on the Cragg-Donald statistic/Kleibergen-Paap rk statistic.

Election year	1979	1983	1990	1994	1999	2002
2nd stage:						
$\beta_2^{\hat{S}LS}$: low- & medium skills	0.358^{***} (0.100)	0.255^{***} (0.063)	0.375^{***} (0.082)	0.328^{***} (0.073)	0.513^{***} (0.084)	0.411^{***} (0.056)
$\beta_2^{\hat{S}LS}$: high skills	1.038 (0.744)	-0.043 (0.494)	-0.897^{*} (0.525)	-1.457^{***} (0.421)	-1.153^{***} (0.404)	-0.217 (0.247)
F-statistics on excl. instruments ^{<i>a</i>}	364/47	364/47	213/47	213/47	83/50	83/50

Table 5. The effect of immigration by education on FPÖ votes: 2SLS cross-section estimations

This table summarizes the estimated effect of immigration (share of residents without Austrian citizenship with low, medium and high education) on the share of votes for the FPÖ based on a series of weighted (community population weights) instrumental variable estimations using Austrian community level data. The dependent variable is equal to the share of votes for the FPÖ in the general election in community *i* in the year *t*, where $i = \{1, \ldots, 2, 352\}$ and $t = \{1979, 1983, 1990, 1994, 1999, 2002\}$. The endogenous variables — for which estimated coefficients and standard errors from the 2nd stage are listed — are the shares of immigrants with low/medium and high education in community *i* in that year, which are instrumented by the respective shares of immigrants in community *i* in the year 1971. Details on the calculation of the share of inhabitants by educational attainment are provided in the notes to Table 1. All regressions control for the community's number of inhabitants, number of inhabitants squared, the distribution of marital status (shares of inhabitants who are single, married, divorced or widowed), the whole age-sex-distribution (34 groups), and binary variables indicating, communities in Vienna, and Carinthia. Robust standard errors (allowing for heterosekdasticity of unknown form) are in parentheses. *, ** and *** indicate statistical significance at the 10-percent level, 5-percent level, and 1-percent level, respectively. ^a Angrist-Pischke multivariate *F*-test of excluded instruments.

Empirical model	С	(1) ard (2007)	Cc	(2) ortes (2006)	(3) Borjas (2006)		
Dependent variable Explanatory variable	$(N_t - N_t) = (F_t - F_{t-1})$	$(N_{t-1} + F_{t-1})/(N_{t-1} + F_{t-1})/(N_{t-1} + F_{t-1})$		$\frac{ln(N_t)}{ln(F_t)}$	$\frac{(N_t - N_{t-1})/\bar{N}}{F_t/(N_t + F_t)}$		
Displacement if Attraction if	$egin{array}{l} eta_{Card} < 0 \ eta_{Card} > 0 \end{array} \ eta_{Card} > 0 \end{array}$		β_{0} β_{0}	Cortes < 0 Cortes > 0	$\begin{array}{l} \beta_{Borjas} < 0 \\ \beta_{Borjas} > 0 \end{array}$		
$\hat{\beta}(s.e.)$ Number of observations	0.051	(0.032) 7,056	0.003	(0.002) 9,408	-0.081	(0.043) 7,056	

Table 6. Empirical models for identifying the internal migration response

This table summarizes estimation output of empirical models for identifying the internal migration response as discussed and evaluated by Peri and Sparber (2011) (henceforth PS). The estimations are based on Austrian community-level panel data with i communities over t years, where $i = \{1, \ldots, 2, 352\}$ and $t = \{1971, 1981, 1991, 2001\}$. The specifications are equivalent to a subset of specifications presented in Table 7 of PS. Each specification controls for community and year fixed effects. Specification (1) is equal to the preferred specification of PS — a slightly modified specification of Card (2001, 2007) — which they describe/recommend on page 90. Specification (2) is denoted by PS as the 'Cortes (2006) alternative', and specification (3) is called the 'Borjas (2006) alternative'. Standard errors are in parentheses. *, ** and *** indicate statistical significance at the 10-percent level, 5-percent level, and 1-percent level, respectively.

Election year	Pooled	Δ_{79-71}	Δ_{83-71}	Δ_{90-71}	Δ_{94-71}	Δ_{99-71}	Δ_{02-71}
Panel A: Immigration 2nd stage:							
$\beta_2^{2SLS^D}$	1.434^{**} (0.620)	$\begin{array}{c} 0.542 \\ (0.361) \end{array}$	2.581^{**} (1.064)	1.916^{*} (1.003)	2.592^{**} (1.177)	1.412^{**} (0.663)	1.306^{**} (0.626)
F-statistics on excl. instrument ^a	149/7	32/7	32/7	23/5	23/5	24/6	24/6
Average change in FPÖ votes Average change in share of immigrants	$9.28 \\ 3.64$	$\begin{array}{c} 0.61 \\ 1.03 \end{array}$	$-0.47 \\ 1.03$	$ \begin{array}{r} 11.18 \\ 3.81 \end{array} $	$17.05 \\ 3.81$	$\begin{array}{c} 21.46 \\ 6.02 \end{array}$	$\begin{array}{c} 4.56 \\ 6.02 \end{array}$
Panel B: Immigration by education							
$\beta_2^{\hat{I}V^D}$: with low and medium skills	1.283^{**} (0.539)	$0.294 \\ (0.192)$	1.688^{***} (0.508)	1.551^{**} (0.718)	2.125^{**} (0.853)	$2.099 \\ (1.395)$	$1.595 \\ (1.058)$
$\beta_2^{\hat{IV}^D}$: with high skills	$3.270 \\ (2.146)$	-0.452 (0.287)	1.527^{*} (0.866)	$2.664 \\ (1.625)$	3.918^{*} (2.329)	$\begin{array}{c} 6.721 \\ (5.619) \end{array}$	$\begin{array}{c} 4.191 \\ (4.332) \end{array}$
$F\mbox{-}{\rm statistics}$ on excl. instruments b	12/12	14/28	14/28	10/15	10/15	4/6	4/6

Table 7. The effect of immigration on FPÖ votes: 2SLS estimations in differences

This table summarizes the estimated effect of the change in the share of immigrants (share of residents without Austrian citizenship) on the change in the share of votes for the FPÖ based on a series of weighted 2SLS estimations using Austrian community level data. In the by-year regressions, the dependent variable is equal to the absolute change in the share of votes for the FPÖ in the general election in community *i* in the year *t* compared to 1971, where $i = \{1, \ldots, 2, 352\}$ and $t = \{1979, 1983, 1990, 1994, 1999, 2002\}$. The pooled regression pools these vote share changes. In Panel A, the endogenous variables — for which estimated coefficients and standard errors from the 2nd stage are listed — are the absolute changes in the share of immigrants in community *i* in the year 1971. In Panel B, the endogenous variables are the absolute changes in the shares of immigrants with low/medium and high education in community *i* in the year *t* compared to the year 1971, which are instrumented by the respective shares of immigrants in community *i* in the year 1971. In Panel B, the endogenous variables are the absolute changes in the shares of immigrants in community *i* in the year 1971. In Panel B, the endogenous variables are the absolute changes of the respective shares of immigrants in community *i* in the year 1971. In Panel B, the endogenous variables are the absolute changes of the distribution of marital status (shares of inhabitants who are single, married, divorced or widowed), the whole age-sex-distribution (34 groups), and binary variables indicating, communities in Vienna, and Carinthia. Each estimation is based on 2, 352 observations. Robust standard errors (allowing for heterosekdasticity of unknown form) are in parentheses. *, ** and *** indicate statistical significance at the 10-percent level, 5-percent level, and 1-percent level, respectively. *a* Wald *F*-statistics based on the Cragg-Donald statistic/Kleibergen-Paap *rk* statistic. *b* Angrist-Pischke multivariate *F*-test of excluded instru

Please note: The following supplementary appendices are not meant for publication in print. It can be made available on a Journal website and the authors' websites upon publication.

Supplementary Appendix A. Additional graphs

The spatial distribution of FPÖ votes over time

Figure A.1 shows the spatial distribution of the share of votes for the FPÖ in the six general elections under consideration. In line with Figure A.2 we see that the share of votes for the FPÖ increases between 1979 and 1999, and drops in 2002. With the exception of a very strong base of support for the FPÖ in the state of Carinthia (located in the south of Austria) no other particular geographical patterns (over time) are evident.

[Insert Figure A.1 here]

Time series of FPÖ vote shares and immigrants in Austria

Austria has witnessed several waves of mass (labor) immigration, which increased the share of immigrants (i. e., residents without Austrian citizenship), shown on the right axis in Figure A.2, dramatically over time.

[Insert Figure A.2 here]



Figure A.1. The spatial distribution of the share of votes for the FPÖ in general elections

These figures depict the share of votes for the FPÖ in Austrian general elections on a community-level for the following election years 1979, 1983, 1990, 1994, 1999, and 2002. The number of communities and their territorial boundaries has changed over the sample period. In order to have a balanced panel of communities, a slightly modified version of the territorial boundaries of the year 2001 with 2,352 communities (including the 23 municipal districts of Vienna) is used.



The share of votes for the *Freedom Party of Austria* (FPÖ) is from parliamentary elections. After 2005 this figure also includes the votes for the newly-established *Alliance for the Future of Austria* (BZÖ), a splinter from the FPÖ. Share of immigrants captures the share of residents without Austrian citizenship. Own calculations based on data from *Statistics Austria*.

Supplementary Appendix B. Additional Tables

Table B.1 shows estimation output with all controls. Table B.2 summarizes robustness to inclusion of control variables. These results are commented in the text.

[Insert Table B.1 here]

[Insert Table B.2 here]

Table B.1.	Full estimation or	utput for the 2SLS	estimation b	based on the	pooled sample

	First stag	ie:	Second st	age:
	Share of immigr	ants in t	Share of FPO	O votes
Immigration				
Share of immigrants in 1971	0.873***	(0.047)		
Share of immigrants	01010	(0.011)	0.386^{***}	(0.045)
Community a			0.000	(010-00)
No. of inhabitants	0.002	(0.001)	-0.001	(0.001)
(No. of inhabitants) ²	-0.001^{***}	(0.001)	0.001	(0.001)
Vienna	0.020**	(0.008)	-0.024^{***}	(0.005)
Carinthia	-0.007^{**}	(0.003)	0.109***	(0.005)
Family status ^b		(0.000)	0.200	(0.000)
Share of singles	0.084^{***}	(0.029)	0.075^{***}	(0.025)
Share of divorced	0.534***	(0.079)	-0.115**	(0.058)
Share of widows	-0.000	(0.041)	0.045	(0.040)
$A a e-sex-distribution^c$		(0.0)	0.0.00	(010-00)
Share of females between 0 and 5	0.046	(0.201)	0.758^{***}	(0.188)
Share of females between 5 and 10	-0.777^{***}	(0.225)	0.628***	(0.170)
Share of females between 10 and 1	5 -0.757***	(0.206)	0.528***	(0.153)
Share of females between 15 and 2	0 -0.692***	(0.177)	0.423***	(0.153)
Share of females between 20 and 2	5 -0.228	(0.205)	0.454^{***}	(0.171)
Share of females between 25 and 3	0 -0.173	(0.203)	0.357**	(0.171)
Share of females between 30 and 3	$5 -0.674^{***}$	(0.223)	0.724***	(0.171)
Share of females between 35 and 4	$0 -1.351^{***}$	(0.234)	1.190***	(0.160)
Share of females between 40 and 4	$5 -1.585^{***}$	(0.345)	1.365^{***}	(0.180)
Share of females between 45 and 5	0 -1.120***	(0.260)	1.723***	(0.179)
Share of females between 50 and 5	$5 -1.090^{***}$	(0.310)	1.078***	(0.193)
Share of females between 55 and 6	0 -0.337*	(0.174)	0.980***	(0.168)
Share of females between 60 and 6	5 -0.267	(0.213)	0.803***	(0.190)
Share of females between 65 and 7	0 -0.303	(0.202)	0.523***	(0.184)
Share of females between 70 and 7	$5 -0.422^*$	(0.255)	0.941***	(0.188)
Share of females between 75 and 8	0 -0.004	(0.224)	-0.478^{***}	(0.183)
Share of females 80 or older	0.047	(0.031)	-0.080^{**}	(0.037)
Share of males between 5 and 10	-0.698^{***}	(0.216)	0.577***	(0.169)
Share of males between 10 and 15	-0.870^{***}	(0.189)	0.546^{***}	(0.143)
Share of males between 15 and 20	-0.524^{***}	(0.180)	0.312^{**}	(0.142)
Share of males between 20 and 25	-0.228	(0.166)	0.287^{**}	(0.146)
Share of males between 25 and 30	0.118	(0.201)	0.040	(0.150)
Share of males between 30 and 35	0.282	(0.203)	0.002	(0.165)
Share of males between 35 and 40	0.450^{*}	(0.231)	0.336^{**}	(0.152)
Share of males between 40 and 45	0.627^{***}	(0.214)	-0.391^{***}	(0.149)
Share of males between 45 and 50	0.458^{*}	(0.245)	0.291	(0.185)
Share of males between 50 and 55	0.150	(0.202)	-0.677^{***}	(0.171)
Share of males between 55 and 60	-0.045	(0.191)	-0.614^{***}	(0.183)
Share of males between 60 and 65	-1.341^{***}	(0.223)	-0.290	(0.195)
Share of males between 65 and 70	-1.656^{***}	(0.239)	0.402^{*}	(0.230)
Share of males between 70 and 75	-1.440^{***}	(0.258)	1.148***	(0.227)
Share of males between 75 and 80	-1.193^{***}	(0.270)	1.320***	(0.281)
Share of males 80 or older	0.075^{*}	(0.043)	-0.027	(0.052)
Election $years^d$	yes	. ,	yes	. ,

The estimations presented provide the full estimation output for the first specification summarized in Table 2 and the third specification summarized in Table $3.^a$ Number of inhabitants is measured in $10.000.^b$ Base group: Share of married ^c Base group: Share of males between 0 and 5 ^d Binary indicators for the election years 1983, 1990,1994, 1999 and 2002. Base group: 1981.

Election year	Pooled	1979	1983	1990	1994	1999	2002
2nd stages:							
No controls	0.270^{***}	0.416^{***}	0.232^{***}	0.342^{***}	0.278^{***}	0.250^{***}	0.102***
	(0.043)	(0.075)	(0.043)	(0.057)	(0.041)	(0.050)	(0.037)
Baseline	0.386^{***}	0.431***	0.230***	0.270***	0.176^{***}	0.338***	0.357***
	(0.045)	(0.063)	(0.040)	(0.065)	(0.059)	(0.063)	(0.042)
+Educational attainment	0.453***	0.392***	0.230***	0.338***	0.271***	0.441***	0.382***
	(0.043)	(0.060)	(0.037)	(0.057)	(0.054)	(0.060)	(0.038)
+Labor market status	0.439***	0.366***	0.209***	0.341***	0.281***	0.466***	0.395***
	(0.041)	(0.060)	(0.037)	(0.057)	(0.055)	(0.061)	(0.039)

 Table B.2.
 Robustness to further controls

The estimations presented in the second row of this table are equal to those presented in Table 3 (third column) and Table 4 in the paper. The estimations in the first row exclude all control variables. The estimations presented in the third row control (compared to the baseline specification) also for the distribution of educational attainment of the total resident population 25 years of age or older; which is captured by the share with no degree, with an apprenticeship, with a lower secondary school, with a higher secondary school or with an academic degree. The estimations presented in the fourth row control (compared to the third row) also for the distribution of labor market status of the total resident population, which is captured by the share which is employed, unemployed, retired; a child or belongs to a residual category.

Supplementary Appendix C. Survey results

We employ data on Austrian respondents from the *European and World Values Survey* (E/WVS).¹ In the years 1990 and 1999 Austrian respondents were asked the question 'If there were a national election tomorrow, for which party on this list would you vote?'

[Insert Table C.1 here]

Table C.1 compares the resulting distribution of stated voting plan among parties in the survey with the actual voting results in the elections closely following the survey dates. We distinguish between *Sample* 1 which includes all respondents who answered the question on their voting behavior and *Sample* 2 which includes only the respondents who provided all the information we use in our subsequent estimation analysis. The results are quite similar for the two samples.

For both years, the survey significantly underestimates the actual vote share that the FPO obtained. The difference is particularly pronounced in 1999: According to the E/WVS, we would have expected about 20 percent of FPÖ voters, whereas in the election the FPÖ scored almost 27 percent of the votes. This finding is consistent with the idea that many voters do not honestly declare that in the voting booth they are voting for an extreme party. (Given the timeliness of the survey poll, it is unlikely that the FPÖ managed to mobilize and/or gain voters to such a great extent in the run-up to the election.)

Bearing the limitations of survey data in mind, we next consider the correlates of preferences for the FPÖ. We construct a binary variable, which is equal to one if a respondent answers 'FPÖ', to the above question and zero otherwise. We then run probit regressions of this variable on a set of demographic variables as well as variables capturing more specifically attitudes toward immigration. Tables C.2 and C.3 contain the results.

[Insert Tables C.2 and C.3 here]

In Table C.2, we find that, by and large, younger, male, less educated, and unemployed individuals as well as those out of the labor force are more likely to have a preference for the FPÖ. Table C.3 demonstrates that several facets of attitudes toward immigrants are strongly associated with voting preferences. For example, those who prefer that scarce jobs are given to native citizens or who even want a complete labor immigration stop are more likely to be in favor of the FPÖ, as are those who do not care about the living conditions of immigrants or are not willing to do something to improve these conditions. These results are broadly consistent with the findings of Lubbers, Gijsberts, and Scheepers (2002) in their analysis of extreme rightwing parties in Western Europe. By contrast, Mayda (2006) and O'Rourke and Sinnott (2006) find that the old are more anti-immigrant than the young. It is difficult to directly compare these studies due to partially different controls.

¹The E/WVS is an academic project organized as a network of social scientists coordinated by a central body, the *World Values Survey Association*. The survey provides data from representative national samples (based on face-to-face interviews) of more than 80 countries. To date, four waves have been conducted: in 1981-1984, 1990-1993, 1995-1997, and 1999-2004.

	Election results Oct 7, 1990	Survey data Apr to J	a (E/WVS) Jun, 1990	Election results Oct 3, 1999	Survey data Aug to (a (E/WVS) Oct, 1999	
		Sample 1 (N=1,052)	$\begin{array}{c} \text{Sample 2} \\ \text{(N=1,014)} \end{array}$		Sample 1 (N=1,041)	Sample 2 (N=888)	
$\mathrm{FP}\ddot{\mathrm{O}}^a$	16.60	15.30	15.09	26.90	20.27	20.20	
$\mathrm{SP}\ddot{\mathrm{O}}^{b}$	42.80	43.25	44.08	33.20	33.05	34.96	
ÖVP ^c	32.10	32.70	32.25	26.90	31.32	30.53	
$\mathrm{GR}\ddot{\mathrm{U}}\mathrm{NE}^d$	6.80	8.56	8.38	7.40	9.70	9.31	
Other	1.70	0.19	0.20	5.60	5.67	4.99	
	100.00	100.00	100.00	100.00	100.00	100.00	

Table C.1. Comparison of survey polls with election results

^a Freiheitliche Partei Österreichs (Freedom Party of Austria) ^bSozialdemokratische Partei Österreichs (Social Democratic Party of Austria) ^c Österreichische Volkspartei (Austrian People's Party) ^dDie Grünen – Die Grüne Alternative (The Greens – The Green Alternative; also called the Austrian Green Party).

	1990		1999a		1999b		1999c		1990 & 19)99a
Age	-0.001	(0.001)	-0.003***	(0.001)	-0.003***	(0.001)	-0.003***	(0.001)	-0.002^{***}	(0.001)
Female	-0.049^{**}	(0.025)	-0.078^{***}	(0.027)	-0.083^{***}	(0.027)	-0.082^{***}	(0.027)	-0.062^{***}	(0.018)
Married	-0.026	(0.026)	0.026	(0.030)	0.028	(0.030)	0.030	(0.030)	0.000	(0.020)
No. of children	-0.007	(0.009)	-0.004	(0.011)	-0.002	(0.011)	-0.001	(0.011)	-0.006	(0.007)
School leaving age	-0.005	(0.003)	-0.009^{**}	(0.004)	-0.009^{**}	(0.004)	-0.009^{**}	(0.004)	-0.008^{***}	(0.003)
Household income	0.003	(0.005)	0.002	(0.005)	0.002	(0.005)	0.002	(0.005)	0.002	(0.004)
Self-employed	0.099^{*}	(0.054)	-0.064	(0.051)	-0.060	(0.052)	-0.058	(0.052)	0.036	(0.039)
Unemployed	-0.005	(0.082)	0.197^{*}	(0.101)	0.215^{**}	(0.104)	0.206^{**}	(0.103)	0.102	(0.070)
Out of labor force	-0.025	(0.031)	0.067^{*}	(0.036)	0.071^{**}	(0.036)	0.066^{*}	(0.036)	0.016	(0.023)
Town, $2,001 - 5,000$	0.057	(0.040)	0.038	(0.042)	0.040	(0.042)	0.034	(0.042)	0.047	(0.029)
Town, $5,001 - 50,000$	0.035	(0.038)	0.052	(0.043)	0.054	(0.043)	0.052	(0.044)	0.037	(0.028)
Town, $> 50,000$	0.099^{**}	(0.048)	0.060	(0.054)	0.062	(0.054)	0.058	(0.053)	0.081^{**}	(0.036)
Austrian citizen		· · · ·		· /	0.120^{**}	(0.060)	0.114^{*}	(0.064)		· · · ·
Interview in 09/99							0.045	(0.031)		
Interview in $08/99$							0.101^{*}	(0.055)		
Year is 1999								· /	0.038^{**}	(0.017)
Federal state FE	Yes		Yes		Yes		Yes		Yes	. ,
Observations	1,014		942		939		939		1,956	
Pseudo \mathbb{R}^2	0.058		0.069		0.072		0.077		0.054	

Table C.2. Determinants of claimed preference for FPÖ, part I

Estimations based on survey data from the E/WVS. The dependent variable is equal to one if an individual stated that she or he would vote for the FPÖ and zero otherwise. The table reports marginal effects. Robust standard errors are in parentheses. *, ** and *** indicate statistical significance at the 10-percent, 5-percent , and 1-percent levels, respectively.

	1990		1999		1990 & 1999	
Would <u>not like</u> to have as neighbors:						
People of different race	0.044	(0.047)	0.106^{*}	(0.056)	0.073**	(0.037)
Muslims	0.004	(0.032)	0.144^{***}	(0.041)	0.072^{***}	(0.027)
Immigrants	0.008	(0.028)	0.201^{***}	(0.046)	0.084^{***}	(0.026)
Observations	1,014		939		1,956	
When jobs are scarce, en	nployers sh	ould give	priority to na	atives over	· immigrants.	
	0.055^{**}	(0.025)	0.140***	(0.024)	0.094^{***}	(0.018)
Observations	967	967 936 1,906		6		
<u>Not at all</u> concerned abo	out the livin	ng conditio	ons of immig	rants.		
			0.137***	(0.026)		
Observations			923			
Absolutely not prepared	to actually	v do somet	hing to impr	rove the co	onditions of imi	migrants.
			0.186***	(0.058)		
Observations			923			
	ibit poople	from othe	er countries o	coming her	re to work.	
Government should proh	ibit people					
Government should proh	ibit people		0.190***	(0.067)		

Table C.3. Determinants of claimed preference for FPÖ, part II

Estimations based on survey data from the E/WVS. The dependent variable is equal to one if an individual stated that she or he would vote for the FPÖ and zero otherwise. The table reports marginal effects. Robust standard errors are in parentheses. *, ** and *** indicate statistical significance at the 10-percent, 5-percent , and 1-percent levels, respectively.